Author's response to reviews

Title: Physical activity, sedentary behaviors, and estimated insulin sensitivity and secretion in pregnant and non-pregnant women

Authors:

Anna Gradmark (anna.gradmark@medicin.umu.se)
Jeremy Pomeroy (pomeroyj@niddk.nih.gov)
Frida Renström (frenstro@hsph.harvard.edu)
Susanne Steiginga (s_steiginga@hotmail.com)
Margareta Persson (margareta.persson@nurs.umu.se)
Antony Wright (Antony.Wright@mrc-epid.cam.ac.uk)
Les Bluck (Les.Bluck@mrc-hnr.cam.ac.uk)
Magnus Domellöf (magnus.domellof@pediatri.umu.se)
Steven E Kahn (skahn@u.washington.edu)
Ingrid Mogren (ingrid.mogren@obgyn.umu.se)
Paul W Franks (paul.franks@med.lu.se)

Version: 2 Date: 18 February 2011

Author's response to reviews:

There were changes made in the manuscript just before submission that hadn’t been updated in the abstract, therefore we revised the abstract.